Accelerator Status Report

Run Schedule: March 11, 2010 to August 13, 2010
August 11, 2010
Injector    Energy 61.88 MeV
· The injector has been slowly drifting. Sometimes Hall B attenuator gets railed to deliver the requested current.
· The QE has been getting low (< 0.2).

· Gun2 was hi-potted to 130 kV to eliminate the field emission.

· (8/2) Reza optimized the injector for QWeak high current run.

· (7/31) Injector drifted a few times. Some minor adjustments corrected the problem.

· The Gun HV does not trip any more after some work on the oil tank.

Accelerator   Linac Energy 550.00 MeV
· (8/10) Zone 1L02 Klystron cooling water leak (1.7 hrs).

· (8/10) iocsl4 crate controller failed (2.5 hrs).
· (8/10) PSS BCM in the BSY dropped the BSY to power permit due to a failed power supply (1.4 hrs).

· (8/8) VB21L17A would not open. It was found that the insulating vacuum pump for zone tripped off and the solenoid for the valve was leaking air. (9.0 hrs)

· (8/8) The CHL tripped (5.5 hrs).

· (8/7) NL RF gang phase was not stable periodically and the instability stopped on its own a few days later. (EES did disconnect a directional coupler and replace a tripler, which did not stop the instability completely.)
· Overall the accelerator was running well and the optics was stable. Some test plans were conducted because the halls were not ready to take beam.
· (8/3, 8/4) NL linac gang phase shifted twice affecting beam. The shift was about 18 degrees. EES is investigating the problem. Zones 1L07 and 1L08 were bypassed and beam was set up to BSY to assist the troubleshooting of the problem.
· (7/30) JT valve for 1L21 was not functional due to problems with the actuator. It was fixed.

· (7/29) Last Thursday the storm dropped the injector to restricted access and the NL RF.

· (7/29) The FSD card for RF 0L04 was replaced. The problem was that the RF trips of the zone did not pull P1faults. The problem has been fixed. 

Hall A---Pass 2, APEX
· Installation of DVCS.

· (7/12) Hall A started the summer down.
Hall B ---Pass 2, g9b-Frost from March 11 to August 13, 2010
· Hall B beam was fine.
· Hall B was happy with its beam.

· (7/29) A few hours were spent to adjust beam profile. M. Tiefenback left three of Hall B

            quads off hysteresis.  

Hall C--- Pass 1, QWeak from July 4 to August 13, 2010
· Continued to commission Qweak and it went well so far.

· (8/9) Tested Hall C raster FSD with beam and it works. When CW beam is higher than 10uA the raster is not maskable.

· Hall C raster power supply may not be stable. The hall will investigate.

·  (8/2) The hall has been down due to a leak in its cryo target. The hall will not take beam until Thursday. 

· Commissioning of the beam line continues and the problems (vacuum, raster, cooling and etc.) were identified and are being addressed. We have not thread beam through the Compton chicane yet. There are problems with the Compton power supply. Hall C is planning to commission its Moller system this week. There is a test plan for this.
· Over the weekend Hall C took beam up to 135uA. 

